Author's response to reviews

Title: Prevalence and associated factors in burnout and psychological morbidity among substance misuse professionals

Authors:

Adenekan Oyefeso (sgju980@sgul.ac.uk)
Carmel Clancy (cclancy@sgul.ac.uk)
Roger Farmer (Roger.Farmer@swlstg-tr.nhs.uk)

Version: 4 Date: 7 February 2008

Author's response to reviews:

Reviewer's report
Title: Prevalence and associated factors in burnout and psychological morbidity among substance misuse professionals
Version: 3 Date: 7 January 2008
Reviewer: Patrick Bovier

Reviewer's report:
The paper has been much improved!

I would just suggest two minor points:
1. All models are adjusted for occupational groups, but this is not mentioned in the models, p. 13. I suggest that you add it.

The manuscript has been revised as advised. (Page 13 paragraph 1)

2. In addition, social workers were at higher risk of emotional exhaustion, but you did not mention in the discussion.

We have added a sentence on page 15 (paragraph 1) in response to this comment